Author's response to reviews

Title: Developing a decision aid to guide public sector health policy decisions: A study protocol

Authors:

Peggy Tso (Peggy.tso@utoronto.ca)
Anthony J Culyer (tony.culyer@utoronto.ca)
Melissa Brouwers (mbrouwer@mcmaster.ca)
Mark J Dobrow (mark.dobrow@utoronto.ca)

Version: 6 Date: 18 April 2011

Author's response to reviews: see over
April 14, 2011

Implementation Science Editorial Team

To whom it may concern,

Following revisions made according to the copy-edited version of the manuscript, we have submitted a revised draft of our manuscript entitled ‘Developing a decision aid to guide public sector health policy decisions: A study protocol’. Only the following copy-edit revision was not retained, for which the explanation follows:

1) In the background paragraph of the abstract the sentence with the following revision was made to the copy-edited version of the manuscript:
   ‘With few studies in the literature offering decision guidance specifically to health policymakers, the present study aims to facilitate the structured and systematic incorporation of research evidence and—where there is currently very little guidance, values, and other non-research-based evidence—into the policy making process.’

   Our intent is not to say that there is little guidance, values and other non-research-based evidence for the incorporation of research evidence. Rather, it is to say that there is little guidance for the incorporation of values and other non-research-based evidence.

We thank you for your assistance in improving this paper.

Yours sincerely,

Peggy Tso, BHSc, MSc Candidate
Department of Health Policy Management & Evaluation
University of Toronto

Anthony J Culyer, CBE, BA, Hon DEcon, Hon FRCP, FRSA, FMedSci
Professor
Department of Health Policy, Management and Evaluation
University of Toronto

Melissa Brouwers, BSc, MA, PhD
Associate Professor
Department of Oncology
McMaster University

Mark Dobrow, MSc, PhD
Assistant Professor
Department of Health Policy, Management and Evaluation
University of Toronto